

JOURNAL OF SPATIAL INFORMATION SCIENCE Number 30 (2025), pp. 61–92

RESEARCH ARTICLE

Network evolution analysis: a usage-centric approach applied on cycling infrastructure

Raphaël Bres^{1,2}, Arnaud Le-Guilcher², Veronika Peralta¹, Ana-Maria Olteanu Raimond², Cyril de Runz¹, and Thomas Devogele¹

¹ LIFAT, Université de Tours, France

² UMR LASTIG, IGN-ENSG, Université Gustave Eiffel, France

Received: June 14, 2024; returned: September 16, 2024; revised: November 27, 2024; accepted: March 31, 2025.

Abstract: Cycling practice is quickly increasing around the world, giving rise to the development of devoted infrastructure to protect its users and offer them a more enjoyable ride. Mobility infrastructure is represented in geographical databases, but these databases are often centered on car and pedestrian mobility. This causes some data quality problems like the lack of completeness or freshness. Volunteered geographical information (VGI) is affected by this kind of problem with a variable extent relying on the contributors' wish and skills. Research on VGI evolution for a network mainly focuses on the main usage of a road section, ignoring secondary information related to other road users of a specific section. This paper contains two contributions. To model the evolution, we define a multiplex graph where each layer represents a snapshot. It is implemented with an infrastructure class based on how cyclists perceive an infrastructure. We also present two complementary VGI road network evolution methods with a usage-centric approach on cycling. These approaches are adaptable for any usage of the network and are based on the multiplex graph. The first approach is based on the road sections, analyzing the evolution of each section individually. The second approach is based on randomly generated starting/ending points. These methods are illustrated in the Centre-Val de Loire region with OpenStreetMap.

Keywords: openstreetmap, cycling network, data evolution, multiplex graph, route generation

1 Introduction

Cycling mobility has experienced a significant increase in popularity since the early 21^{st} century [31]. This trend has been further accelerated by the COVID-19 pandemic, as individuals sought alternatives to public transportation. Cycling is recognized as an environmentally friendly alternative to car travel, offering the advantage of shorter routes in urban areas. Additionally, cycling for commuting, daily activities or leisure contributes to improved human health and reduced CO_2 emissions. In response to this growing popularity, public authorities have been implementing new cycling infrastructure solutions to enhance its mobility and ensure traffic safety by protecting cyclists from motor vehicles. Consequently, numerous cycling infrastructure have been developed, predominantly in urban areas. However, these infrastructure remain limited in many rural areas, where higher car speeds are common. According to [6], in Austria, rural mobility administrators continue to prioritize car infrastructure over cycling mobility.

Cyclists typically utilize roads recommended by route planning applications. However, these applications can only provide recommendations if cycleways are present and accurately documented in road databases. Updating cycling infrastructure in road network databases in near real-time poses a significant challenge for national mapping agencies and volunteered geographical information (VGI) platforms such as OpenStreetMap (OSM), due to the rapid nature of infrastructure changes. These changes often involve modifications to existing infrastructure rather than the creation of entirely new elements (e.g., adding a new roundabout takes longer than modifying an existing infrastructure to create a new cycleway), necessitating local information and prompt updates.

In response to the increase in both cycleways and cycling mobility, the French national mapping agency revised its topographic road network model (BDTOPO[®]) at the end of 2023 to include diverse cycling infrastructure. However, incorporating that information into the database on a national scale is both costly and time-consuming. Conversely, while VGI platforms such as OSM have extensive coverage of the cycling network [12], maintaining quality and continuous updates remains challenging. This is because the editing activity relies on the motivation, skills, and local knowledge of contributors, leading to heterogeneities between rural and urban areas. These local differences result in variable data quality within the dataset. The only study on a high-quality dataset specifically for cycling has been conducted solely for Paris [40].

These data quality issues have inspired numerous research studies focusing on data quality and data evolution for VGI [3, 14, 16]. A primary limitation of these studies is the lack of a usage perspective, as roads within a network are typically analyzed only for their primary users (e.g., major roads for cars, tracks for pedestrians). For instance, a major 2x2 lane straight road may be highly efficient for motor vehicles due to the high speeds achievable, but cyclists often find such roads dangerous and tend to avoid them [17]. A usage-centric analysis of the road network is more beneficial for optimizing its use, particularly in specific routing scenarios.

For cycling, ensuring optimal conditions involves considering numerous factors, as demonstrated in [20]. These factors include safety, the landscape attractiveness, and the efficiency of reaching the destination, which are challenging to balance. Due to the slower development of cycling infrastructure, [25] noted that OSM data in rural areas are not updated as quickly as in urban areas, resulting in better OSM data quality in major cities compared to rural regions.

This paper examines data evolution in a road network from a usage perspective. It proposes two complementary approaches based on (i) road sections and (ii) generated routes. Both approaches utilize a classification system applied to each road section, reflecting how cyclists perceive road infrastructure. The first approach involves analyzing each OSM road section individually to determine its cycling infrastructure class. The second approach involves generating shortest path routes with a priority on using cycling infrastructure. A way centrality indicator is then computed for each road section to estimate the use of cycling infrastructure in the generated routes. Additionally, the distribution of infrastructure classes is analyzed for each route. A multiplex graph is created and instantiated to model network evolution over time [11]. These approaches are applied to the Centre-Val de Loire region where many rural areas exist as well as some important cities.

This paper presents three key contributions:

- 1. The development of a multiplex graph to model the evolution of road networks over time.
- 2. The introduction of two complementary approaches for usage-centric VGI evolution: a road section approach focused on individual road sections, and a routes approach centered on cycling network usage. Both approaches are based on the multiplex graph.
- A case study applying both approaches for cyclists in the Centre-Val de Loire region. These contributions are generic and can be adapted to other uses, such as for cars or individuals with reduced mobility, provided the infrastructure classes are appropriately modified.

The paper is structured as follows: Section 2 reviews the state of the art in OSM data evolution and multiplex graphs. Section 3 introduces the multiplex graph and the proposed approaches. Section 4 describes the study area and the parameters for both approaches. Section 5 presents the results and Section 6 discusses them. Section 7 concludes the paper and suggests directions for future research.

2 Related work

This section reviews related work on the definition and representation of cycling networks. It then discusses methods for analyzing the evolution of these networks. Additionally, it examines studies on evolution graphs to provide a robust and user-friendly tool for our data evolution approaches.

2.1 The cycling network: definition and representation

The linear component of the cycling network, as defined in [4], encompasses all roads where cyclists are permitted. This includes dedicated cycling infrastructure such as cycleways and cycle lanes, as well as portions of the pedestrian network, and various roads without specific cycling infrastructure. A subset that includes only dedicated cycling infrastructure [36,37] would result in a fragmented network impractical for route generation in most countries. This definition is more suited to urban planners. For our purposes, a usage-centric definition of the cycling network, as proposed in [4], is required. That definition includes every road section where cyclists are allowed. In both definitions, the cycling network consists of points representing intersections and lines representing roads.

This subsection also examines and discusses various models for the cycling network. The cycling network is expanded to include other elements in [2], such as points of interest and road signalization, which are not linear. The cycling network has been modeled in urban planners' databases, such as Vélo & Territoire's database¹. Vélo & Territoire is an association that assists local authorities in developing their cycling infrastructure and provides data models to store information about all aspects of the cycling network. Their roads database for cycling is detailed at the road section level, with attributes to differentiate the right and left sides of the road. This structure attracted the French mapping agency, which partially incorporated it into BDTOPO[®] for the road network. Only roads with cycling infrastructure (cycleways, cycle lanes, roads shared with other users) are represented in it. Additionally, Vélo & Territoire has modeled other components of the cycling network as defined in [2]. However, this representation is not formalized and does not accommodate all road users (e.g., pedestrians, motor vehicles).

Cycling infrastructure were previously implemented in an oriented graph by [26]. In this study, the road network is represented by an oriented graph, with vertices representing network intersections and edges representing road sections. Each edge is associated with multiple costs, including time and security costs, for each direction. This graph was implemented with a single snapshot, but its structure could be adapted to accommodate multiple snapshots. A multi-layer graph, described by [9], incorporates multiple snapshots, providing a list of snapshots where an edge or a vertex is available. However, this graph consumes excessive memory because identical objects (edge or vertex) are recreated in multiple snapshots. A potential cost reduction strategy involves using the same object (edge or vertex) across all layers of the graph.

We recognize that numerous research efforts have focused on correlating high-quality cycling infrastructure with increased bicycle usage [7]. Also, cycling level of traffic stress [13] shows how cycling policies have been oriented mainly towards the separation from other usage, which is not a way to represent every type of cyclist. There are multiple routing engines proposing cycling options with OSM data like r5r [29], OSRM [22], openrouteservice ² or Valhalla ³. They do include some bike dedicated roads and some cars/pedestrian dedicated roads. However, there remains an ongoing challenge concerning the usage patterns of all roads accessible to cyclists in the road network databases, regardless of their designated cycling infrastructure.

The linear component of the cycling network has been defined; however, the state of the art reveals the existence of multiple definitions [2, 4, 37]. Additionally, various representations of the cycling network exist (e.g., Vélo & Territoires, [9]), but these are either not formalized or too costly for long term data evolution analysis. The next subsection discusses how data evolution has been studied.

2.2 Spatial data evolution

Spatial data evolution is defined as the transformation of a spatial database over time, encompassing all dimensions of data quality and the evolution of the database structure to address an expanding range of problems. Within the context of VGI, our primary focus

¹https://www.velo-territoires.org/politiques-cyclables/data-velo-modeles-donnees/

schema-donnees-amenagements-cyclables/

²https://maps.openrouteservice.org/

³https://github.com/valhalla/valhalla?tab=readme-ov-file

is on the data quality aspect of spatial data evolution. Research on spatial data evolution has predominantly centered on OSM, particularly in relation to quality assessments [14,19, 23,43]. Investigating OSM data quality raises critical questions regarding completeness, freshness, and other relevant quality metrics.

In this paper, we focus on data freshness that we define as the temporal aspects of data [28]. Numerous studies [14, 16] have examined data freshness in the context of network evolution, positing that increased temporal dynamics correlate with a higher number of contributors and daily usage, which may enhance both completeness and freshness. Conversely, [19,23] focus their evolution analysis exclusively from a completeness perspective, concentrating on the quantitative changes of individual road. In contrast, this paper adopts a perspective that emphasizes the analysis of evolution in relation to specific usage rather than solely through a quality lens.

From our perspective, network evolution analysis is exemplified by the work of [24], which examines the global evolution of network length categorized by road type. In this study, we will similarly investigate variations in length over time

The volume of evolution studies pertaining to OSM data is on the rise, facilitated by the Geofabrik website and the OSHDB format [32]. The Geofabrik website provides access to snapshots of OSM data across various global regions at multiple temporal points. The OSHDB format enables users to retrieve either a snapshot of a region for a specific date or the complete historical dataset for an area spanning two dates. This framework has been employed in prior research, as illustrated in [41].

Consequently, research on data evolution can be categorized into two distinct approaches: contribution-based analysis and snapshot-based analysis. These approaches may focus on different subjects of analysis; some studies examine individual road segments, while others assess networks from a usage perspective. We refer to these methodologies as feature-centric and usage-centric, respectively.

The contribution-based analysis utilizes comprehensive historical OSM data for a given region, examining each contribution to derive results. In the feature-centric approach, [25] investigated contributions across multiple cities worldwide, categorizing them based on whether they involved OSM points of interest (POI) additions, deletions, geometric updates, or attribute updates. The differentiation between types of updates, specifically attribute and geometric updates, represents a significant and emerging area of research, as highlighted by [25]. In this context, we will propose the introduction of a third category of updates, focusing on attribute modifications that result in changes to the cycling usage of any OSM way. It is important to note that [25] analysis was conducted on a dataset of POI, which is not the primary focus of our study.

To the best of our knowledge, no contribution-based analysis has been conducted employing a usage-centric approach.

A snapshot analysis involves obtaining multiple states of OSM data for a specific area and aggregating the changes that occur between two consecutive snapshots. Several studies have employed this method using a feature-centric approach across various countries [24,41,43]. These studies utilize different time intervals between their snapshots and timeframes for analysis. For instance, [24] employed a three month interval over a study period of 4.5 years, while [43] used the same interval within a four year study period. However, the network usage aspect that we aim to incorporate into our analysis renders these short time intervals and limited study durations inadequate. Furthermore, these studies do not differentiate between usage patterns in their approaches to road networks.

Only two applications of snapshot methods utilizing a usage-centric approach have been identified: [33], which focuses on car usage, and [5], which examines bike usage. Both studies employ randomly generated routes to evaluate network usage. The primary aim of [33] and [5] is to compare the lengths of these routes. However, [5] additionally proposes various environments to assess the method and metrics based on the number of routes for which the length was altered by at least 1% and 20%. This study further elucidates the rationale behind specific changes and introduces the concepts of *update*, defined as a real-world change promptly reflected in OSM, and *upgrade*, characterized as a resolution to a data quality issue.

Data evolution can be categorized into two primary methods: snapshot-based and contribution-based approaches. These methods can further be classified as either featurecentric or usage-centric. Usage-centric approaches are particularly relevant to our research objective, as they focus on the perception of features from the perspective of cyclists, ensuring that each of them is analyzed based on how it is experienced by users. To facilitate the analysis of cycling network data evolution, a tool is required to store and manage multiple snapshots of OSM data.

2.3 Evolution graphs

This subsection presents various graphs that illustrate the evolution of networks. Given that road networks are commonly modeled as graphs, several extended graph-based data structures have been proposed to represent the dynamic nature of evolving networks. We define an *edge* as the line component of a graph, representing the connections between points, and a *vertex* as the point component of a graph, representing the intersections within the network. We reserve the terms *node* and *way* for their specific meanings in the OSM context, where a node represents a specific geographic point, and a way represents a sequence of nodes forming a polyline, such as a road.

In the field of spatio-temporal dynamic graphs, [10] and [27] review several spatiotemporal graph models used to represent urban networks. The model proposed by [30] challenges the conventional approach by representing streets as vertices and intersections as edges. However, a key limitation of this model is the absence of attribute properties necessary to distinguish road segments based on specific usage. Additionally, this representation is not well suited for route generation. The study also provides an extensive analysis of the concept of centrality to support the graph's structure.

Several graph models have been developed with a focus on the semantic web. [15] introduced a graph model that integrates temporal information into the RDF format, a standard defined by the W3C. In this work, two methods for incorporating the temporal dimension into RDF graphs are proposed: the timestamp approach and the snapshot approach, both closely related to the data evolution methods discussed previously. The result is a triplet extended by two timestamps that define its validity period, enabling the creation of snapshots at any point in time.

Temporal property graphs, such as those described by [18], have also been defined. These models combine the RDF format with a property graph, which allows the storage of attribute information for vertices. The temporal property feature enables the handling of time and the storage of multiple attribute values, as well as vertices and edges versioning. This aspect of versioning is particularly useful for managing OSM data, as the majority of its objects have multiple versions and numerous tags that can be treated as attributes. However, a limitation of this approach is the authors' failure to specify whether each object in the graph must implement an attribute, as is the case in OSM. Additionally, [39] proposed a duration labeled graph where each edge is associated with a start date and a validity period, offering a low computational cost for query processing.

Temporal dynamic graphs have also been applied in other fields, such as traffic prediction, as demonstrated in [34] and [38]. In these studies, temporal dynamic graphs are integrated into neural networks to forecast traffic patterns. However, these approaches fall outside the scope of this research and are therefore not considered in our analysis.

To the best of our knowledge, no existing research has proposed a graph structure that incorporates multiple snapshots with semantic attributes and is optimized for efficiently handling multiple OSM snapshots. The structure introduced by [30] is closely aligned with our needs, but additional attribute properties are required to integrate the detailed information available for each road section in OSM. The multi-layer approach proposed by [9] is also essential for capturing data evolution, although we aim to reduce the computational cost associated with such a graph.

A multiplex graph, as described by [8], which integrates attribute properties and temporal dimensions, presents the most suitable option for our study. It offers an effective balance between ease of query processing and low memory usage. The specifics of our graph model and data evolution methods are detailed in Section 3.

3 Methods and tools to analyze the cycling network evolution

This section presents our approach to analyze the evolution of the cycling network. The overall pipeline is presented in Figure 1.



Figure 1: Pipeline to evaluate the cycling network.

We begin by acquiring OSM data from Geofabrik. Subsequently, we classify the infrastructure based on OSM tags. This classification is then used to instantiate the graph, with each infrastructure element assigned a corresponding class. During graph instantiation, both the road sections and routes-based approaches are implemented, and results are collected for each. The road section approach is performed prior to the routes-based approach to facilitate the selection of areas of interest (AoI) for the latter method.

In this section, we also present the multiplex graph model and describe the data evolution methods used in our analysis.

3.1 Multiplex graph for VGI road network evolution

As outlined at the conclusion of Section 2, we utilize a multiplex graph G to analyze the OSM cycling network. This graph is designed with multiple interconnected layers to accommodate the various data snapshots, each identified by the parameter n which is our temporal parameter in this article. Attributes are assigned to both the edges and vertices, with greater emphasis on the edges. This emphasis reflects the primary focus of our analysis, which is centered on road sections rather than intersections.

Let $G^n = (E^n, V^n)$, where G^n represents a graph layer corresponding to the snapshot at timestamp *n*. E^n denotes the set of all edges for snapshot *n*. Each edge in the graph corresponds to a road section in the real world, where its characteristics remain uniform throughout the entire segment. Equation 1 illustrates the composition of E^n .

$$E^{n} = \{e_{0}^{m}, e_{1}^{m}, ..., e_{i}^{m}, e_{k}^{m}\}$$

$$\tag{1}$$

In Equation 1, e_i^m represents an edge with version m in snapshot n of the graph, where k denotes the total number of edges in snapshot n.

 V^n represents the set of all vertices in the graph for snapshot *n*. Each vertex corresponds either to an intersection or a point where there is a change in the attributes of the road. The composition of V^n is presented in Equation 2.

$$V^{n} = \{v_{0}^{m}, v_{1}^{m}, ..., v_{j}^{m}, v_{l}^{m}\}$$

$$(2)$$

In Equation 2, v_j^m represents a vertex with version m in snapshot n, where l denotes the total number of vertices in snapshot n.

$$G^{n} = (\{e_{1}^{m}, e_{2}^{m}, ..., e_{i}^{m}, ..., e_{k}^{m}\}, \{v_{1}^{m}, v_{2}^{m}, ..., v_{i}^{m}, ..., v_{l}^{m}\})$$
(3)

As shown in Equation 3, the union of E^n and V^n constitutes a single layer of the multiplex graph. Each graph layer G^n , corresponding to a snapshot, collectively forms the full multiplex graph G. The union of all E^n and V^n across snapshots creates the complete set of edges E and vertices V, respectively. For edges and vertices that remain unchanged between two consecutive snapshots, they are applied to both layers without duplication.

Edges in *E* correspond to ways in the OSM database that contain the "highway" key and permit cyclist access. Each edge is characterized by a linear geometry, an identifier (id), a version number (version) to its evolution, an OSM id (osmid), an OSM version (osmversion) to trace the source way in OSM, a cycling infrastructure classification (cyc) to monitor changes in infrastructure, and a field consolidating all OSM tags associated with the way (fields) to track attribute changes.

🗧 www.josis.org

Vertices in V correspond to OSM nodes that either define the endpoints of ways or indicate a change in attributes within a road section in E. Each vertex v_i^m is assigned a unique identifier (id), a version number (version), an OSM id (osmid), an OSM version (osmversion), and point type geometry (geom). Since the primary focus of this study is not on the vertices, no cycling-related attributes are included for them.

Each edge is bounded by two vertices. The relationship between an edge and its corresponding vertices is defined in Equation 4.

$$e_i^m = \overrightarrow{v_j, v_{j'}} \tag{4}$$

The graph versioning process is partially based on the versioning system used in OSM. To explain how versioning is handled in our graph, the various update scenarios are outlined as follows.

In the first case, if an OSM way remains unchanged between two consecutive snapshots, the corresponding edge in the graph is left unmodified. The second case involves geometric changes to an OSM way, which can occur in two forms: either as a geometric change to a node or as a change in the composition of nodes within the way. A geometric change in a node occurs when the geometry of a node in the way is altered, affecting all ways that use that node. In OSM, such changes are visible only at the node level. In our graph, this change is reflected in both the vertex corresponding to the node and all edges that are connected to it. This scenario highlights why it is insufficient to rely solely on OSM versioning for our graph. A node composition change, on the other hand, occurs when there is a change in the set of nodes that make up the way. This could involve the addition, removal, or modification of nodes.

The third case occurs when an OSM way experiences a version update due to an attribute modification. If multiple modifications (whether geometric, attribute-related, or both) occur between two consecutive snapshots, the changes are grouped together, and the version number in the graph is incremented by only one. This versioning logic is applied even if an edge or vertex undergoes both geometric and attribute changes. Similarly, in the fourth case, when attributes are added, removed, or modified, the edge or vertex's version number is updated. If several attribute changes occur between two snapshots, they are grouped together, and the version number is incremented by one. This same logic is applied in cases where multiple types of changes occur between snapshots. Additionally, when a new OSM way is created, it always starts with version one in the graph, even if it has been modified in OSM between its creation and the first snapshot.

If an OSM way is deleted and had been modified between the last snapshot and its deletion, a new version representing the final state of the edge is not created.

Figure 2 provides an example of how the graph operates, with the infrastructure classes in this example corresponding to the categories presented in Table 4.2.

Multiple methods have been developed to facilitate access to specific properties of the edges:

- The geometry is represented as $g(e_i^m) \rightarrow e_i^m g$.
- The attribute information is denoted by $t(e_i^m) \rightarrow e_i^m t$, which encompasses all tags associated with an OSM way.
- The infrastructure information is represented as $c(e_i^m) > e_i^m . c.$
- The length is indicated by $d(e_i^m) \rightarrow e_i^m d$.

It is important to note that all methods described above are also applicable to the vertices, with the exception of the length.

Figure 2 shows the three considered layers in this example. The first layer at timestamp



Figure 2: Example of three snapshots of the same graph.

n is composed of $V^n = \{v_1^1, v_2^1, v_3^1, v_4^1, v_5^1, v_6^1\}$ and $E^n = \{e_1^1, e_2^1, e_3^1, e_4^1, e_5^1, e_6^1\}$.

Having established the framework for the n layer of the graph, it is important to note that several changes have occurred in the data between timestamps n and n + 1, resulting in the formation of the n + 1 layer.

- A new vertex v_7^1 and a new edge e_7^1 have been created.
- Vertex v_6^1 has undergone a geometric change, which subsequently altered the geometry of edge e_5^1 . As a result, v_6^1 is now represented as v_6^2 and e_5^1 is represented as e_5^2 .
- Edge e_1^1 has experienced an infrastructure change, as it has now been designated as cycling-prioritized. Consequently, e_1^1 is now denoted as e_1^2 .
- Edge e_3^1 has undergone a geometric change without any modifications to its vertices, and is now represented as e_3^2 .
- Edge e¹₆ has experienced an attribute change, although its infrastructure class remains unchanged. It is now denoted as e²₆.

These changes are represented as written in Equation 5 and 6.

$$V^{n+1} = V^n \cup \{v_7^1, v_6^2\} - \{v_6^1\}$$
(5)

www.josis.org

$$E^{n+1} = E^n \cup \{e_7^1, e_1^2, e_3^2, e_5^2, e_6^2\} - \{e_1^1, e_3^1, e_5^1, e_6^1\}$$
(6)

Significant changes have occurred between the n + 1 and n + 2 snapshots, as detailed below:

- Vertex v_8^1 has been created in the location where a bridge previously separated edges e_1^2 and e_5^2 . The bridge has now been demolished, and an intersection has been established in its place.
- Edges e_{10}^1 and e_{11}^1 have been introduced to replace edge e_6^2 . Both of these new edges reflect an infrastructure change, as they now prioritize bicycle transport instead of pedestrian access.
- Edge e₁² has undergone a geometric modification to accommodate the newly established intersection at vertex v₈¹, resulting in its designation as e₁³.
- Vertex v_9^1 and edge e_{12}^1 have also been created.
- Edge e_3^2 and vertex v_4^1 have been deleted from the graph.

These changes are represented as written in Equation 7 and 8.

$$V^{n+2} = V^{n+1} \cup \{v_8^1, v_9^1\} - \{v_4^1\}$$
(7)

$$E^{n+2} = E^{n+1} \cup \{e_8^1, e_9^1, e_{10}^1, e_{11}^1, e_{12}^1, e_1^3\} - \{e_2^1, e_3^2, e_6^2, e_1^3\}$$
(8)

Following the explanation of the three layers that contribute to the definitions of E and V, Equations 9 and 10 represent these elements across each layer.

$$E = E^n \cup E^{n+1} \cup E^{n+2} = \{e_1^1, e_1^2, e_1^3, e_2^1, e_3^1, e_2^2, e_4^1, e_5^1, e_5^2, e_6^1, e_6^2, e_7^1, e_8^1, e_{10}^1, e_{11}^1, e_{12}^1\}$$
(9)

$$V = V^{n} \cup V^{n+1} \cup V^{n+2} = \{v_1^{1}, v_2^{1}, v_3^{1}, v_4^{1}, v_5^{1}, v_6^{1}, v_6^{2}, v_7^{1}, v_8^{1}, v_9^{1}\}$$
(10)

This multiplex graph will be utilized to implement the approaches discussed in the subsequent subsection.

3.2 Analyzing data evolution in the cycling network

This subsection presents a study divided into two parts, each focusing on distinct aspects of the data. The first part, termed the *road sections approach*, examines road sections across the entire study area, while the second part, referred to as the *routes approach*, is based on routes generated within smaller segments of the network known as AoI. Both approaches utilize information derived from the previously described graph.

The road sections approach involves a quantitative analysis of the evolution of the cycling network at the edge level. Each edge is considered individually to assess the evolution of its characteristics, thereby conducting a comprehensive evaluation of all edges within the cycling network. This approach is widely employed and deemed sufficient for analyzing extensive areas. Specifically, it considers the number of added, deleted, and updated edges, categorizing updates into three types: geometric, attribute, and infrastructure. Geometric updates pertain to changes in position and shape for an edge, while attribute updates encompass any modifications to an edge attribute. Infrastructure updates involve any changes that result in the evolution of an edge cycling infrastructure label.

The maps generated through the road sections approach serve to identify representative AoI within the study area, capturing the various patterns present. However, it is important to note that the road section approach assigns equal weight to all road sections or more precisely, to each kilometer regardless of the significance of a particular road section for cyclists. Consequently, we also conduct a routes approach, which aims to provide a more usage-centric analysis of the network and its evolution within the identified AoI.

Before delving into the routes approach, several definitions are necessary. A route is defined as an ordered sequence of road sections (for example OSM ways) connecting a starting point to an ending point within a specified set of interconnected roads, referred to as a network. Let p denote the count of routes for a given AoI r and layer n. The notation $I_{p,r}^n$ represents the route corresponding to the start-end pair p for snapshot n within area r. It is characterized by a linear geometry. Let $comp(I_{p,r}^n)$ denote the composition of the ordered road sections utilized by the route.

The routes approach, primarily based on [5], involves a quantitative analysis of the evolution of the simulated cycling network usage. Simulated usage is achieved by computing optimal routes between various start-end point pairs. These routes are analyzed over time, meaning that optimal routes are calculated for each snapshot. For this analysis, each route corresponding to each start-end point pair is examined individually, focusing on its geometric characteristics and the road sections involved. This paper builds upon the methodology presented in reference [5], which generates random starting and ending points to create a route for each pair across all available layers in the graph.

With the graph and approaches thoroughly described, the subsequent section will detail the experiments conducted.

4 Experiments

This section will present the study area and the implementation of both approaches presented above.

4.1 Study area

The Centre-Val de Loire region is situated in the central part of France (see Figure 3). This region is characterized by several medium-sized cities, such as Tours and Orléans, along-side numerous smaller towns, including Blois, Bourges, and Chartres. Predominantly rural, the region features significant agricultural areas, particularly in Beauce, as well as extensive forested regions, notably in Sologne.

The Centre-Val de Loire region boasts numerous tourist attractions, notably the Loire châteaux, including Chambord, Cheverny, and Chenonceau. In recent years, the region has promoted slow tourism through the development of a network of cycling routes that connect key tourist destinations while prioritizing safe travel. Central to this initiative is 'The Loire by Bike' ⁴, which is part of EuroVelo, the European cycling network. Figure 4.1 illustrates the cycling network as defined in Section 2. It is important to note that the cycling routes are not depicted in Figure 4.1 due to their classification as low traffic roads that lack dedicated cycling infrastructure.

In the study area, a significant majority of roads are part of the motor vehicle network, as illustrated in Figure 4.1 (left). While cyclists have access to these road infrastructure, they do not provide specific enhancements to facilitate cycling use and are generally not

⁴https://www.loirebybike.co.uk/homepage/la-loire-a-velo-nature-culture-and-adventure/



Figure 3: Area of study [5].

designed with cyclist convenience in mind. Consequently, cyclists must share the same roadway space with motorized vehicles. The major cities within the region are identifiable by the presence of dedicated pedestrian (indicated in blue) and cyclist (indicated in green) pathways, as shown in Figure 4.1 (left).

Figure 4.1 (right) further illustrates that green mobility initiatives are concentrated along the Loire river. The primary urban centers in this area include Orléans, Blois, and Tours; however, the remaining cycling infrastructure in the region are relatively sparse.

4.2 Implementation

In this subsection, the methodologies for implementing the road section approach and the routes approach are described in detail.

The road section approach employs various indicators that focus on the number of additions, deletions, and updates of road sections, as well as the length of the road sections affected by these operations. Updates to road sections are categorized into three types: geometric updates, infrastructure updates, and attribute updates.



Figure 4: OSM cycling network in the area of study by classes: the left map including infrastructure dedicated to the motor vehicles and forbidden roads, and the right map without them.

A geometric update refers to changes in the position or shape of a road section and is represented in our code by the boolean variable *geomChange*. This indicator is necessary because, in OSM, a positional change in a node results in a positional change for every way associated with that node, yet there is no corresponding update to the version number of the way in OSM. In our graph, the version number must be incremented in such cases. The *geomChange* variable is set to true when the geometry of the edge differs between two consecutive snapshots, which is determined by the test $g(e_i^m) \neq g(e_i^{m+1})$, where $e_i^{m+1} \in E^n$ and $e_i^{m+1} \in E^{n+1}$.

Within the category of attribute updates, infrastructure updates are specifically highlighted. This is indicated by the boolean variable *infrastructureChange*, which is set to true when the cycling infrastructure class differs between two consecutive snapshots, as determined by the test $c(e_i^{m+1}) \neq c(e_i^{m+1})$, where $e_i^{m+1} \in E^n$ and $e_i^{m+1} \in E^{n+1}$.

Among the attribute updates, infrastructure updates are specifically identified and represented by the boolean variable *infrastructureChange*. This variable is assigned a value of true when there is a difference in the cycling infrastructure class between two consecutive snapshots. This condition is evaluated using the test $c(e_i^m) \neq c(e_i^{m+1})$, where $e_i^m \in E^n$ and $e_i^{m+1} \in E^{n+1}$.

The cycling infrastructure classes are detailed in Table 4.2, which presents two distinct classifications. The first classification, referred to as *infrastructure families*, groups multiple infrastructure types into a single category based on a shared prioritized user. The second classification, termed *infrastructure classes*, provides a more granular classification of cycling infrastructure. It is noteworthy that, in certain instances, the inclusion of prohibited roads

is essential for understanding the evolution of the network. These roads will be considered in this study when deemed necessary.

Family	Class	Definition	OSM definition
Cycling infrastructure	Cycleway	Road section dedicated to bikes where cars are mostly forbidden except for local use	highway = cycleway
	Cycle lane	Space dedicated to bikes on a road section	<pre>cycleway or cycleway:both or cycleway:right or cy- cleway:left = {lane, oppo- site_lane}</pre>
	Shared road	Road section where bikes and cars, and sometimes pedestrians, share the same space together	{cycleway = {shared_lane, opposite}} or {oneway = no and lanes = 1 and cycle- way = lane} or {cyclestreet = yes}
	Shared busway	Road section dedicated to buses where cyclists are al- lowed	{cycleway or cycle- way:both or cycle- way:right} and cycle- way:left = share_busway} for the main direction of travel, add oneway:bicycle = no for the use of the opposite direction
Pedestrian infrastructure	Pedestrian ways	Road section where pedes- trians are the only users al- lowed. A cyclist can use these roads only if he walks alongside his bike	highway = {pedestrian, bridleway, footway, path, sidewalk, crossing, liv- ing_street}
	Stairs	Particular pedestrian in- frastructure because they are more difficult to access when pushing or carrying a bike	highway = steps
Infrastructure dedicated to the motor vehicles	Infrastructure dedicated to the motor vehicles	Road sections where the cars are prioritized and nothing exists for bikes	Any OSM way having the key highway that is not in- cluded in another class
Forbidden roads	Forbidden roads	Roads forbidden to cyclists	highway = {motorway, motorway_link, trunk, trunk_link} or bicycle = no

Table 1: Cycling network classifications.

For attribute changes, any modification made to an edge is considered an evolution. In the context of OSM, if any tag associated with a feature is added, deleted, or modified, the edge or vertex is deemed to have undergone an attribute change. This condition is represented by the boolean variable *attributeChange*, which is set to true when the OSM tags of an edge differ between two consecutive snapshots. This relationship is determined using the test $t(e_i^m) \neq t(e_i^{m+1})$, where $e_i^m \in E^n$ and $e_i^{m+1} \in E^{n+1}$.

Subsequently, indicators of cycling network evolution are computed based on two consecutive layers of the graph. These indicators include the total length of added edges, denoted as $d(E_{add}^{n+1}) = d(E_{tot}^{n+1})/d(E_{tot}^n)$, the total length of deleted edges, defined as $d(E_{del}^{n+1}) = d(E_{tot}^{n+1})/d(E_{tot}^n)$, and the total length of modified edges. The updates are categorized into three types: geometric updates, which pertain to changes in geometry; attribute updates, which refer to changes in any properties of the edge; and infrastructure updates. It is important to note that an update may fall into multiple categories simultaneously. All of these indicators are weighted by the length of each edge.

Regarding the routes approach, the parameters of the experiment are in accordance with [5]. For each AoI and each year, a total of 1,000 routes are generated. The starting and ending points are fixed for the same AoI across different years. A starting or ending point is initially generated within the AoI and mapped to the network. The point is retained only if it remains consistent across all temporal snapshots and is located within 100 meters of the network. The Euclidean distance between the starting and ending points ranges from 300 meters to 5 kilometers. An additional output has been integrated into this approach, which is a way centrality indicator (*cent*) calculated at the level of a road section version. This indicator, computed on edges, is similar to the way centrality as described by reference [42]. However, our method differs in that the shortest path algorithm has been adapted to identify routes that are more suitable for cycling. Specifically, we employ Dijkstra's algorithm but modify the weights of the road sections to favor cycling-friendly roads, such as cycleways.

In our analysis, we compute the way centrality indicator for a single step, utilizing the result for subsequent analyses. Let x represent the number of routes utilizing an edge e_n^i for a given year n, and let y denote the total number of calculated routes within an AoI for that year. The formula is given by $cent(e_n^i) = x$, where $x \in [0, y]$. Using this indicator, we assess the evolution of the mean, the 99th percentile, and the maximum values for each AoI and for each snapshot. Road segments with a zero way centrality indicator are excluded from the analysis to focus on road sections that have been utilized at least once within a single AoI and snapshot.

Another component of the routes approach involves the assessment of cycling infrastructure, for which we define two safety levels. The most stringent level considers only cycleways, as they provide complete separation for cyclists from other road users. The second level encompasses the family of cycling infrastructure, which includes cycle lanes designated bike-specific spaces as well as shared roadways with cars, where motorized vehicles can also operate, thus presenting a greater risk. Additionally, shared roadways with buses, while typically wide and low-traffic, pose potential challenges during interactions with buses.

Indicators have been developed to calculate the number of routes that utilize at least one cycleway and the number of routes in which cycleways comprise at least 30% of the total route length. Having outlined the experimental framework, Section 5 will present and discuss the results obtained from this study.

76

5 Results

This section presents the results derived from both approaches, accompanied by an indepth discussion. The first subsection focuses on the road section analysis, where various results are examined and interpreted, including detailed commentary on specific roads. To ensure accuracy, these changes have been verified through manual investigations using Google Street View.

5.1 Road section analysis

The cycling network comprises various types of roadways, including those exclusively designated for cycling, such as cycleways, as well as shared spaces where cyclists coexist with other road users, including cars, buses, and pedestrians. The initial step in our study involves presenting the evolution of the infrastructure families discussed in Section 4. Table 2 illustrates the changes in road length across these three categories, with the percentages reflecting the evolution from the snapshot at time *n* compared to that at time n - 1.

Year	Cycling infras-	Pedestrian	Motor vehicles in-	Total	
Teal	tructure	infrastructure frastructure		10141	
2014	828.4	2602.2	92960.2	96390.8	
2015	932.7 (+12.6%)	3227.7 (+24%)	109207.3 (+17.5%)	113367.7 (+17.6%)	
2016	1086.7 (+16.5%)	3820.9 (+18.4%)	114158.8 (+4.5%)	119066.5 (+5%)	
2017	1204.5 (+10.8%)	4228.7 (+10.7%)	120545.6 (+5.6%)	125978.7 (+5.8%)	
2018	1209.1 (+0.4%)	6331.0 (+49.7%)	129185.6 (+7.2%)	136725.7 (+8.5%)	
2019	1373.6 (+13.6%)	7092.8 (+12%)	133515.9 (+3.4%)	141983.4 (+3.8%)	
2020	1370.0 (-0.3%)	7556.4 (+6.5%)	138191.7 (+3.5%)	147118.1 (+3.6%)	
2021	1526.2 (+11.4%)	8047.2 (+6.5%)	140364.0 (+1.6%)	149937.4 (+1.9%)	
2022	1623.2 (+6.4%)	8766.3 (+8.9%)	143278.3 (+2.1%)	153667.8 (+2.5%)	

Table 2: Length in km per year of road sections according to the cycling family.

As anticipated, infrastructure dedicated to motor vehicles constitutes the majority of the road network, exhibiting a gradual increase over time. Notably, between 2014 and 2015, there is a significant growth across all types of infrastructure. The length of cycling infrastructure has notably doubled from 2014 to 2022, although there are two periods of stagnation observed in 2018 and 2020. The only decrease in cycling infrastructure length occurred between 2019 and 2020, with a reduction of 0.3%. Conversely, the pedestrian network experienced a substantial increase between 2017 and 2018, marked by a 49.7% rise, likely attributable to a surge in contributions related to sidewalks in major cities. This growth in pedestrian infrastructure length surpasses that of cycling infrastructure during the same period.

Table 3 shows the lengths of road sections belonging to each editing category between snapshots at time n and n + 1. The percentage indicates the ratio of the length of relevant edges and the total length of edges for this snapshot.

Table 4 presents the lengths of updated edges categorized into attribute, infrastructure, and geometric updates. The percentages are relative to the road lengths of the updated edges, as indicated in Table 3. It is noteworthy that an update may encompass both ge-

Period	Added	Deleted	Same	Updated
2014-2015	18405.4 (16.9%)	1428.6 (1.3%)	59035.0 (54.3%)	29843.7 (27.5%)
2015-2016	12550.2 (10.8%)	700.6 (0.6%)	76742.6 (66.1%)	26085.2 (22.5%)
2016-2017	11062.8 (8.9%)	525.2 (0.4%)	87909.2 (70.6%)	25081.7 (20.1%)
2017-2018	13100.6 (9.7%)	497.6 (0.4%)	87305 (64.8%)	33928.3 (25.2%)
2018-2019	8218.2 (5.8%)	561.1 (0.4%)	99916.9 (71.1%)	31850.6 (22.7%)
2019–2020	7685.2 (5.3%)	514.7 (0.4%)	105156.1 (71.8%)	33005.8 (22.6%)
2020-2021	4762.3 (3.2%)	865.5 (0.6%)	116247.0 (77.9%)	27286.7 (18.3%)
2021-2022	5650.8 (3.7%)	474.8 (0.3%)	120891.5 (78.9%)	26201.0 (17.1%)

Table 3: Length in km per year of road sections according to their edition type.

ometric and infrastructure changes simultaneously, which accounts for the row totals not summing to 100%.

Period	Geometric	Infrastructure	Attribute
2014-2015	26187.6 (87.7%)	271.7 (0.9%)	10782.9 (36.1%)
2015-2016	23112.2 (88.6%)	335.1 (1.3%)	7249.8 (27.8%)
2016-2017	22829.9 (91%)	385.1 (1.5%)	5045.6 (20.1%)
2017-2018	29104.6 (85.8%)	229.1 (0.7%)	10061.8 (29.7%)
2018-2019	25246.0 (79.3%)	527.3 (1.7%)	12103.4 (38%)
2019-2020	30696.7 (93%)	292.3 (0.9%)	5511.0 (16.7%)
2020-2021	24798.5 (90.9%)	405.9 (1.5%)	5353.1 (19.6%)
2021-2022	22580.0 (86.2%)	479.2 (1.8%)	6677.2 (25.5%)

Table 4: Length in km per year of updated road sections according to their update type.

The lengths of added and deleted edges tend to decrease when the length of updated edges remains stable. Additionally, the length of updated edges decreases when compared to the total length of the network. In contrast, the length of edges that remain unchanged between two consecutive snapshots, n and n + 1, increases both in absolute terms and as a percentage, exhibiting increases of 54.3% between 2014 and 2015 and 78.9% between 2021 and 2022. Geometric updates constitute the majority of updates in OSM, accounting for 80% to 90% throughout the study period. Attribute updates represent between 25% and 40% of the total updates, with a notable decline towards the end of the study period, recording only 16.7% between 2019 and 2020.

This observation suggests that OSM contributors in the Centre-Val de Loire region are gradually achieving consensus regarding the data quality in this area, indicating that the number of changes driven by data quality issues is not substantial. Consequently, the remaining updates predominantly reflect actual changes in the real world.

Table 5 illustrates the evolution of the lengths of edges across different infrastructure classes between two consecutive snapshots, specifically from 2018 to 2019. The evolution appears stable across the different snapshots, and this particular year serves as a fair representation of how the data evolves over time.

Figure 5 illustrates the usage conflict between cyclists and pedestrians, revealing that 3% of the cycle paths from 2018 are now classified as part of the pedestrian network. This shift is primarily attributed to the incorrect classification of these pathways, which were originally designated as shared spaces for both cyclists and pedestrians. The increasing

2019 2018	Cycleways	Cycle lane	Shared way	Pedestrian	Motor vehicles	Forbidden	Do not exist	Total
Cycle path	90.2%	0.0%	0.0%	3.0%	0.8%	0.0%	6.0%	100%
Cycle lane	0.0%	97.6%	0.6%	0.0%	1.6%	0.0%	0.3%	100%
Shared way	0.0%	0.2%	96.5%	0.1%	3.0%	0.0%	0.3%	100%
Pedestrian	0.2%	0.0%	0.0%	93.4%	4.9%	0.0%	1.5%	100%
Motor vehi-	0.0%	0.0%	0.0%	0.1%	99.6%	0.0%	0.3%	100%
cle	0.078	0.070	0.070	0.170	JJ.070	0.078	0.570	100 /0
Forbidden	0.0%	0.0%	0.0%	0.0%	0.3%	98.1%	1.7%	100%
Do not exist	1.4%	0.3%	0.1%	13.2%	81.8%	3.2%	0.0%	100%
Total	0.5%	0.3%	0.2%	5.0%	91.6%	2.1%	0.4%	100%

Table 5: Evolution of the length of the edges of each infrastructure class between 2018 and 2019.

number of cyclists utilizing these shared infrastructure has created safety concerns for pedestrians, leading to the decision to restrict many of these pathways exclusively for pedestrian use [35]. Additionally, it is noteworthy that a substantial number of pedestrian infrastructure have been established, accounting for 13.2% of all road creations, in contrast to the mere 1.4% for cycleways.

Figure 5 displays the edges categorized by editing type for the periods 2014-2015 and 2021-2022 across three distinct areas: Tours, one of the two major cities; Blois, a medium-sized city; and Sologne, a predominantly forested and rural region.

Figure 5 illustrates that the Sologne area exhibited a denser network in 2022, with numerous road sections remaining unchanged between 2021 and 2022. In contrast, Tours displayed a significant number of updates across both maps, revealing clusters of activity, such as the green cluster on the 2021-2022 map, and red clusters indicating road sections with minimal changes. This pattern may suggest that OSM contributors have reached a consensus regarding those areas. The overall network density remains consistent with the density observed on the 2014-2015 map. Conversely, Blois experienced an extensive campaign between 2021 and 2022 focused on adding sidewalks in the city center, which accounts for the substantial number of new edges added. This has contributed to a greater overall density compared to the 2014-2015 map.

Overall, our findings indicate that the cycling networks in the two major cities, Tours and Orléans, the medium-sized cities (Blois, Bourges, Chartres, Châteauroux), and the rural areas exhibit distinct characteristics. To conduct a detailed analysis of the evolution in these areas, we selected six zones categorized into three groups: the major cities represented by Tours and Orléans, the medium cities comprising Blois and Bourges, and the rural areas including the Sologne forest and Beauce valley. These zones are depicted in Figure 3.

The road section analysis provides general insights into road infrastructure and its evolution within our study area, treating each road section uniformly. However, from a cyclist's perspective, certain road sections warrant prioritization over others. Consequently, the routes analysis serves to complement the road section analysis by focusing on the sections utilized by cycling routes over time.



Figure 5: Change types in Tours, Blois, and Sologne between 2014–2015 and 2021–2022.

5.2 Routes analysis

Firstly, the AoI encompass varying numbers of road sections, as illustrated in Table 6, which depicts the evolution of the number of edges utilized by at least one generated route for each AoI. It is important to note that any road section within an AoI that is not employed in the generated routes is excluded from this table.

Year/AoI	Tours	Orléans	Blois	Bourges	Sologne	Beauce
2014	4058	2433	1467	1342	945	1087
2015	4788 (18.0%)	2466 (1.4%)	1603 (9.3%)	1533 (14.2%)	1032 (9.2%)	1136 (4.5%)
2016	4892 (2.2%)	2588 (4.9%)	1701 (6.1%)	1718 (12.1%)	1098 (6.4%)	1231 (8.4%)
2017	5004 (2.3%)	2832 (9.4%)	1720 (1.1%)	1842 (7.2%)	1122 (2.2%)	1453 (18.0%)
2018	5171 (3.3%)	2857 (0.9%)	1745 (1.5%)	1894 (2.8%)	1178 (5.0%)	1588 (9.3%)
2019	5339 (3.2%)	3191 (11.7%)	2049 (17.4%)	1933 (2.1%)	1310 (11.2%)	1643 (3.5%)
2020	5413 (1.4%)	3646 (14.3%)	2120 (3.5%)	1995 (3.2%)	1372 (4.7%)	1681 (2.3%)
2021	5525 (2.1%)	3906 (7.1%)	2099 (-1.0%)	2081 (4.3%)	1408 (2.6%)	1690 (0.5%)
2022	5679 (2.8%)	4043 (3.5%)	2408 (14.7%)	2215 (6.4%)	1457 (3.5%)	1718 (1.7%)

Table 6: Number and evolution of road sections per year used by at least one generated route for each AoI.

AoI	Tours	Orléans	Blois	Bourges	Beauce	Sologne
Year						0
2014	11135	6272	2661	2650	2023	2244
2015	14977	6978	3406	3652	2539	2701
2016	16658	7868	3919	4428	2890	3327
2017	17874	8934	3978	4849	4052	3591
2018	19076	9317	4161	5059	4703	4244
2019	20917	14267	4849	5410	4969	4757
2020	21861	17738	5450	5798	5125	5288
2021	22899	20324	6497	6194	5243	5585
2022	24793	21689	8931	6973	5411	5776

Table 7: Evolution of the road sections number in the AoI.

In major cities, the number of road sections is significantly higher, indicating that there are numerous routing options available for any given origin-destination pair with minimal impact on routing time. Conversely, rural areas offer fewer optimal options for navigation between origin-destination pairs, resulting in a lower number of utilized road sections.

Regarding the evolution of road sections, its number for the same AoI consistently increases between consecutive snapshots, with the exception of Blois between 2020 and 2021, where the count decreased from 2,120 road sections to 2,099. This decline can be attributed to the rectification of a data quality issue concerning a major road (i.e., Boulevard des Cités Unies), which comprises multiple road sections. This road had been designated as prohibited for bicycles since at least 2005 but had not been accurately represented in OSM prior to this correction.

The overall increase in the number of road sections is of a similar magnitude across all AoI. Notably, Tours experienced a significant increase in its number of road sections between 2014 and 2015, while the other areas exhibited a more linear progression in their growth.

Following the examination of the number of edges involved for each AoI, we proceed to analyze the evolution of the mean way centrality indicator. This analysis is visually represented in Figure 6.

The overall trend of the mean way centrality indicator exhibits a gradual decline over time for each AoI. However, notable local increases are observed between consecutive snapshots, such as the increase in Blois between 2020 and 2021, attributed to the classification of a major road as prohibited for cyclists during that period. It is important to note that this trend does not exhibit significant variation based on the type of AoI. To gain deeper insights into the higher values of way centrality, the analysis of the 99th percentile values for each AoI across the years will be presented.

In Table 8, the 99th percentile of way centrality demonstrates a decreasing trend for Orléans, Bourges, Sologne, and Beauce. The values for Blois and Tours remain stable, although the indicator for Blois exhibits fluctuations over time. Notably, there is considerable variability observed in Bourges, Beauce, and Orléans, with values of 146 for Bourges and 195 for Orléans in 2014, contrasting with 128 for Bourges and 214 for Orléans in 2015.

An unexpected finding is the minimal influence of urbanization on this variable. Whether an AoI is classified as a major city, a small city, or a rural area appears to have little impact on the way centrality values. Conversely, the internal structure of the city is

BRES ET AL.



Figure 6: Evolution of the way centrality mean.

Year/AoI	Tours	Orléans	Blois	Bourges	Sologne	Beauce
2014	37.43	70.68	65	85.59	30	27.14
2015	36.13	71.7	61	87.36	28	26.65
2016	34	71.26	61	72.83	28.03	21.7
2017	34	64	60.81	75.95	25.79	18.96
2018	32	63.44	62	75.28	25.23	16.13
2019	35	65	54	79.68	23	16
2020	35	65.1	54	79	23	16
2021	36	57	62.04	81.2	29.93	16
2022	36	57.58	64	77	22	16

Table 8: Way centrality 99th percentile per year for each AoI.

more significant. For instance, the most frequently utilized roads in Blois are consistently the bridges, which can be attributed to the presence of the Loire river. A major alteration to one of these bridges resulted in a substantial increase in the way centrality indicator between 2020 and 2021.

For Beauce, the way centrality indicator has stabilized at a value of 16 over the past four years, indicating that the majority of the roads in this area have already been established and that the AoI is not experiencing rapid evolution. Additionally, the infrastructure class of the most utilized road sections primarily consists of those dedicated to motor vehicles, with sporadic presence of cycle lanes. Therefore, modifications to the route generator may be necessary to more effectively prioritize cycling infrastructure.

The subsequent section of the results concerning the routes approach focuses on analyzing the proportion of cycling infrastructure incorporated into each generated route over time. Notably, the Beauce AoI was excluded from the results, as it consistently produced 999 out of 1000 routes each year that utilized 100% of the motor vehicle network. Figure 7



illustrates the number of routes that include at least one cycleway over the observed time period.

Figure 7: Number of routes per year using at least one cycleway.

The analysis reveals that urban areas are generally more conducive to cycling, with a significantly greater availability of cycling infrastructure compared to rural regions, where only 26 routes utilized at least one cycleway in 2022. However, there is a pronounced upward trend in the number of routes incorporating a cycleway over time, with the exception of Tours, where the number of routes using at least one cycleway fluctuates and remains relatively constant from the beginning to the end of the study period.

Notably, there is no discernible difference in cycleway usage between large and small cities throughout the study. Even among cities of similar sizes, patterns in cycleway utilization are inconsistent; for instance, Tours shows stagnation across the entire study period, while Orléans exhibits a significant increase in cycleway usage, suggesting an important underlying trend. This increase is particularly evident between 2016 and 2017, as well as between 2020 and 2021.

Figure 8 illustrates the number of routes that incorporate at least one cycling infrastructure over time. The results from this analysis are comparable to those presented in Figure 7.

The distinction between rural areas and urban centers is further emphasized in this analysis. While there has been a marked increase in the utilization of cycleways over time, rural areas continue to exhibit a lower number of routes incorporating cycling infrastructure by the conclusion of the study period. For instance, Tours experienced an increase from 466 to 754 routes, whereas Sologne saw a more modest rise from 2 to 32 routes.

Significant spikes in data are evident in Figure 8, particularly between 2018 and 2019 for Blois and Bourges. However, it remains challenging to ascertain whether these increases are correlated. Further investigation is warranted, particularly focusing on the behavior and editing activities of OSM contributors during that period. Analyzing whether the same contributors made significant contributions in both areas concurrently, or if bulk imports [21] from open data sources occurred, could yield insights.





Figure 8: Number of routes per year using at least one cycling infrastructure.

Overall, the number of routes utilizing at least one cycling infrastructure increased by over 1,000 between the beginning and end of the study period. Notably, the most substantial increases were observed in Bourges and Orléans, suggesting that the scale of a city does not significantly influence the usage of its cycling infrastructure. Although the increases for Blois and Tours are less pronounced, they remain noteworthy. Meanwhile, Sologne's growth in cycling infrastructure usage is gradual, but it still lags considerably behind the levels seen in urban areas.

Following the observation that many routes incorporate only a limited amount of cycling infrastructure, Figure 9 illustrates the evolution of the number of routes in which at least 30 percent of the route comprises cycleways over time.

Initially, it is important to note that Sologne does not exhibit any routes utilizing at least 30% of cycleways. In contrast, the evolution observed in the four cities varies significantly. Tours demonstrates a decrease in routes using cycleways between 2014 and 2016, followed by a modest increase thereafter. The notable decline between 2015 and 2016 can likely be attributed to an infrastructure change in which a cycleway was reclassified as a pedestrian infrastructure. This particular road, segmented into multiple sections, previously had a way centrality indicator ranging from 40 to 50.

Orléans shows a substantial increase in the number of routes incorporating cycleways, particularly following the onset of the COVID-19 crisis. Conversely, Bourges experiences a trend similar to Orléans; however, the COVID-19 pandemic resulted in decreased theoretical utilization of cycleways rather than an increase. This significant reduction for Bourges is primarily linked to a cycleway, which is part of the *Coeur de France à vélo* bike route, being incorrectly tagged as a pedestrian infrastructure over half of the AoI.

In the case of Blois, it was discovered that a bridge was not accurately represented in OSM until the 2016 snapshot. The sidewalks were mistakenly labeled as cycleways, although the actual infrastructure consisted of two sidewalks adjacent to a four-lanes road and two cycle lanes. Given the limited options for crossing the Loire rver, this bridge features prominently in many routes, and the absence of the inaccurately classified cycleway



Figure 9: Number of routes per year using at least 30% of cycleways.

is reflected in the results. Notably, the data for 2021 in Blois presents a minor decrease in the number of routes utilizing at least 30% cycleways between 2020 and 2022, despite 2021 being the year with the highest proportion of routes achieving this threshold. This phenomenon may be attributed to temporary cycleways that were mapped in OSM during the COVID-19 crisis, which were subsequently removed before the 2022 snapshot.



Figure 10: Number of routes per year using at least 30% of cycling infrastructure.

In Figure 10, with the exception of Sologne which consistently exhibits routes with less than 30% cycling infrastructure across all years—each city shows an increasing trend in the number of routes utilizing a higher percentage of cycling infrastructure. By the end of the

study period, Tours and Orléans demonstrate comparable figures; however, Tours had a significant number of routes incorporating cycling infrastructure from the outset.

Orléans began with a number of essential cycleways and gradually expanded its network to include cycle lanes and shared paths over time. The COVID-19 crisis played a pivotal role in this development, with the most substantial growth occurring between 2020 and 2021. Additionally, a notable increase is observed between 2016 and 2017.

Bourges experienced a significant rise in routes using cycling infrastructure between 2018 and 2019, with subsequent years reflecting only modest increases. Blois exhibited a gradual growth in this area from 2018 to 2021. When combined with the data from Figure 9, it can be inferred that cycleways were likely replaced by other types of cycling infrastructure, such as cycle lanes or shared paths, by 2022.

For Tours, the trend is consistent with Figure 9, where a slight decrease is noted at the beginning of the study period, followed by an increase toward the end. This increase in cycling infrastructure is more pronounced than the growth in cycleways.

6 Discussion

The proposed multiplex graph offers an effective solution for analyzing data evolution while minimizing computational costs. Given that OSM snapshots are extensive, even for small areas, managing numerous snapshots can become resource-intensive. Our graph design allows for the incorporation of only the data relevant to the area of study and the selected snapshots, making it operationally efficient.

This multiplex graph facilitates an insightful analysis of OSM data evolution through two complementary approaches. The first is the road section approach, which, although comprehensive for a large area like a French region, may be too broad and diverse to capture every evolutionary pattern effectively. However, this approach enables the visualization of data evolution and provides insights into the trends within the cycling network.

Additionally, the infrastructure classification utilized in the confusion matrix offers valuable information regarding the current trends in an area from a cyclist's perspective. This could pave the way for identifying similar roads in analogous regions that may require updates, potentially utilizing machine learning algorithms to enhance the analysis and application of OSM data.

The application of both approaches provides valuable insights into the evolution of the OSM road network from a cyclist's perspective. The road section analysis reveals that road infrastructure predominantly cater to motor vehicles, particularly outside major cities. Over time, the network has become denser, thanks to the addition of new roads and smaller paths to OSM. This trend aligns with findings in the existing literature on OSM data evolution, which indicates an overall increase in the number of features.

The COVID-19 crisis significantly influenced many cities, leading to the creation of numerous cycleways. This period marked a pivotal shift in urban planning, as cyclists began to be recognized as a legitimate mode of transportation that requires dedicated infrastructure. While motor vehicle infrastructure continues to dominate the network, some roads have been repurposed for pedestrians and cyclists. Additionally, certain routes that were once designated for cyclists have transitioned to pedestrian-only pathways. These changes reflect a broader trend toward slowing down urban environments, as evidenced by many cities reducing speed limits from 50 km/h to 30 km/h to enhance the safety of vulnerable road users.

The road section approach indicates a growing consensus regarding the OSM motor vehicle network, with a similar agreement developing for cycling infrastructure, albeit at a later stage, alongside that for pedestrian pathways. The frequency of updates reflects the ever-changing nature of the real world, underscoring the continuous need for contributions to OSM, a sentiment that aligns with findings from previous research.

In contrast, the routes approach highlights a clear disparity between urban and rural areas. Rural regions often lack sufficient cycling infrastructure, and where it does exist, it tends to cater primarily to tourism around significant landmarks, rarely serving as the fastest routes. To foster cycling as an effective means of transportation across the country, enhancing infrastructure in rural areas is crucial.

While some differences were noted between cities, it remains inconclusive whether a city's size impacts its cycling usage. An important variable absent from this study is the political orientation of city officials over time; ecology-minded decision-makers could potentially correlate with significant increases in cycling infrastructure, observable one or two years following their election.

Another key discussion stemming from the routes approach is the allocation of space between cyclists and pedestrians. When infrastructure for both groups were originally designed, urban planners underestimated the volume of cyclists and their impact on pedestrian safety. The recent surge in cycling has highlighted this issue, as the infrastructure often prove too narrow to accommodate both cyclists and pedestrians safely. Consequently, urban planners have begun redirecting cyclists to the road or implementing measures that slow them down on pedestrian prioritized pathways.

Enhancements to the route generator are also warranted, particularly in weighting cycling infrastructure more heavily and minimizing reliance on major highways. As noted in [1], bikeability encompasses safety, comfort, attractiveness, directness, and coherence. The current routing system tends to prioritize directness over these other important criteria, potentially leading to cyclists navigating dangerous roads intended for motor vehicle traffic.

Moreover, creating different cyclist profiles such as families, commuters, and tourists could introduce diverse routing patterns. This approach mirrors initiatives like the Geovélo route generator, which aims to accommodate the varied purposes cyclists may have for their journeys.

Finally, the observed trend of increasing routes utilizing at least 30% cycling infrastructure signals that more such infrastructure are being developed in the real world. Integrating these facilities into road databases is essential to ensure they are well-known and effectively used by cyclists.

7 Conclusion

In this article, we present a novel approach for analyzing VGI with a focus on user perspectives. This method has been applied to the OSM road network from the viewpoint of cyclists in the Centre-Val de Loire region, covering the years 2014 to 2022. Our analysis reveals the evolving state of OSM data in this region, where data is still in flux, but overall stability is increasingly evident. The primary reason for changes made by OSM contributors in this area is to align the OSM dataset with real-world developments. Notably, our study identifies differences in the development of efficient cycling networks between major cities like Tours and Orléans compared to medium-sized cities such as Blois and Bourges. In rural areas, while cycling infrastructure is lacking, low-traffic roads often substitute for designated cycleways, although they typically offer lower safety.

We have instantiated a multiplex graph to facilitate the analysis of VGI road data. This graph enables the integration of multiple snapshots of any VGI road dataset, with OSM being the most suitable choice for implementation. The methodology we employed is reproducible for any other road users, provided that relevant infrastructure classes can be defined and instantiated.

Numerous future research perspectives exist. For instance, the graph could be enhanced with additional attributes at the vertex level. Integrating road intersections into the analysis is another significant improvement, as intersections are critical points in any navigation system, particularly for cyclists, who face heightened danger at major junctions.

Improvements could also be made to the routes approach. Increasing the weight of cycling infrastructure during the route generation phase would promote its usage. Additionally, starting and ending points for routes could be determined based on GPS data, selected semi-randomly from points of interest or residential areas. This method would reduce major biases associated with network density. The AoI could include diverse settings, such as densely populated regions (like the Paris agglomeration) or mountainous areas, where slope plays a crucial role in routing. Another potential area of investigation involves understanding route changes between two consecutive snapshots for the same starting and ending points. Currently, there is no method for identifying which specific modifications caused a change in the routing. Furthermore, we could develop a bikeability index calculated first on the edges and then on the vertices, utilizing the graph developed by [26] from a routing perspective.

The usage conflicts highlighted in our research offer a pathway for further exploration by adapting our methodology to other use cases, as it is fully generic to any user perspective. We anticipate that by applying this study framework across various usage scenarios, we can identify road sections that are frequently utilized by multiple users, making usage conflicts more pronounced.

Acknowledgments

We greatly thank the Région Centre-Val de Loire and IGN for co-financing this research.

References

- AHMED, T., PIRDAVANI, A., WETS, G., AND JANSSENS, D. Bicycle Infrastructure Design Principles in Urban Bikeability Indices: A Systematic Review. *Sustainability* 16, 6 (2024), 2545. doi:10.3390/su16062545. Number: 6 Publisher: Multidisciplinary Digital Publishing Institute.
- [2] ARANTXA, J. Comparaison des principaux manuels européens d'aménagements cyclables: aménagement cyclable et espace urbain. Tech. rep., Centre d'études sur les réseaux, les transports, l'urbanisme et les constructions publiques (CERTU), 2000.

www.josis.org

- [3] ARSANJANI, J. J., BARRON, C., BAKILLAH, M., AND HELBICH, M. Assessing the Quality of OpenStreetMap Contributors together with their Contributions. In AGILE 2013 (Nashville, TN, USA, 2013).
- [4] BRES, R., PERALTA, V., LE-GUILCHER, A., DEVOGELE, T., OLTEANU-RAIMOND, A.-M., AND DE RUNZ, C. Spécification et qualité du réseau cyclable, application à la recherche d'itinéraires. In *INFORSID* (2022).
- [5] BRES, R., PERALTA, V., LE-GUILCHER, A., DEVOGELE, T., OLTEANU RAIMOND, A.-M., AND DE RUNZ, C. Analysis of cycling network evolution in OpenStreetMap through a data quality prism. In *AGILE: GIScience Series* (Delft, Netherlands, 2023), vol. 4, pp. 1–9. doi:10.5194/agile-giss-4-3-2023.
- [6] BREZINA, T., LEMMERER, H., AND LETH, U. Mental barriers in planning for cycling along the urban–rural gradient. *Transportation Research Interdisciplinary Perspectives* 16 (2022), 100689. doi:10.1016/j.trip.2022.100689.
- [7] BUEHLER, R., AND DILL, J. Bikeway Networks: A Review of Effects on Cycling. *Transport Reviews* 36, 1 (2016), 9–27. doi:10.1080/01441647.2015.1069908.
- [8] CHU, X., FAN, X., YAO, D., ZHU, Z., HUANG, J., AND BI, J. Cross-Network Embedding for Multi-Network Alignment. In *The World Wide Web Conference* (New York, NY, USA, 2019), WWW '19, Association for Computing Machinery, pp. 273–284. doi:10.1145/3308558.3313499.
- [9] COSTES, B., PERRET, J., BUCHER, B., AND GRIBAUDI, M. An aggregated graph to qualify historical spatial networks using temporal patterns detection. In *The 18th AG-ILE International Conference on Geographic Information Science* (Lisbon, Portugal, 2015).
- [10] DEL MONDO, G., STELL, J., CLARAMUNT, C., AND THIBAUD, R. A Graph Model for Spatio-temporal Evolution. J. UCS 16 (2010), 1452–1477.
- [11] DOMENICO, M. D., SOLE-RIBALTA, A., COZZO, E., KIVELA, M., MORENO, Y., PORTER, M. A., GOMEZ, S., AND ARENAS, A. Mathematical Formulation of Multilayer Networks. *Physical Review X* (2013).
- [12] FERSTER, C., FISCHER, J., MANAUGH, J., NELSON, T., AND WINTERS, M. Using openstreetmap to inventory bicycle infrastructure: A comparison with open data from cities. *International Journal of Sustainable Transportation* (2019). doi:10.1080/15568318.2018.1519746.
- [13] FURTH, P., MEKURIA, M., AND NIXON, H. Network connectivity for low-stress bicycling. *Transportation Research Record: Journal of the Transportation Research Board* 2587 (2016), 41–49. doi:10.3141/2587-06.
- [14] GIRRES, J., AND TOUYA, G. Quality Assessment of the French OpenStreetMap Dataset. *Transactions in GIS* 14, 4 (2010), 435–459. doi:10.1111/j.1467-9671.2010.01203.x.
- [15] GUTIERREZ, C., HURTADO, C., AND VAISMAN, A. Introducing Time into RDF. *IEEE Transactions on Knowledge and Data Engineering* 19, 2 (2007), 207–218. doi:10.1109/TKDE.2007.34.

- [16] HAKLAY, M. How Good is Volunteered Geographical Information? A Comparative Study of OpenStreetMap and Ordnance Survey Datasets. *Environment and Planning B: Planning and Design* 37, 4 (2010), 682–703. doi:10.1068/b35097.
- [17] HARDINGHAUS, M., AND PAPANTONIOU, P. Evaluating Cyclists' Route Preferences with Respect to Infrastructure. Sustainability 12, 8 (2020), 3375. doi:10.3390/su12083375. Number: 8 Publisher: Multidisciplinary Digital Publishing Institute.
- [18] HARTIG, O. Reconciliation of RDF* and Property Graphs. Tech. Rep. arXiv:1409.3288, arXiv, 2014. arXiv:1409.3288 [cs]type: article.
- [19] HERFORT, B., LAUTENBACH, S., PORTO DE ALBUQUERQUE, J., ANDERSON, J., AND ZIPF, A. A spatio-temporal analysis investigating completeness and inequalities of global urban building data in OpenStreetMap. *Nature Communications* 14, 1 (2023), 3985. doi:10.1038/s41467-023-39698-6.
- [20] KELLSTEDT, D., SPENGLER, J., FOSTER, M., LEE, C., AND MADDOCK, J. A scoping review of bikeability assessment methods. *Journal of Community Health* 46 (2021), 211– 224. doi:10.1007/s10900-020-00846-4.
- [21] LE GUILCHER, A., OLTEANU-RAIMOND, A.-M., AND BALDE, M. B. ANALYSIS OF MASSIVE IMPORTS OF OPEN DATA IN OPENSTREETMAP DATABASE: A STUDY CASE FOR FRANCE. ISPRS Annals of the Photogrammetry, Remote Sensing and Spatial Information Sciences V-4-2022 (2022), 99–106. doi:10.5194/isprs-annals-V-4-2022-99-2022. Conference Name: XXIV ISPRS Congress "Imaging today, foreseeing tomorrow", Commission IV - 2022 edition, 6–11 June 2022, Nice, France Publisher: Copernicus GmbH.
- [22] LUXEN, D., AND VETTER, C. Real-time routing with openstreetmap data. ACM SIGSPATIAL 2011 (2011), 513–516. doi:10.1145/2093973.2094062.
- [23] MINAEI, M. Evolution, density and completeness of OpenStreetMap road networks in developing countries: The case of Iran. *Applied Geography* 119 (2020), 102246. doi:10.1016/j.apgeog.2020.102246.
- [24] NEIS, P., ZIELSTRA, D., AND ZIPF, A. The Street Network Evolution of Crowdsourced Maps: OpenStreetMap in Germany 2007–2011. *Future Internet* 4, 1 (2011), 1–21. doi:10.3390/fi4010001.
- [25] NOVACK, T., VORBECK, L., AND ZIPF, A. An investigation of the temporality of OpenStreetMap data contribution activities. *Geo-spatial Information Science* (2022), 1– 17. doi:10.1080/10095020.2022.2124127.
- [26] NÉRON, E., RAULT, T., AND REPUSSEAU, F. Bicycle Network Improvements. In ROADEF 2023 (24ème édition du congrès annuel de la Société Française de Recherche Opérationnelle et d'Aide à la Décision) (Rennes, France, 2023).
- [27] OBEROI, K. S., DEL MONDO, G., DUPUIS, Y., AND VASSEUR, P. Spatial Modeling of Urban Road Traffic Using Graph Theory. In *Proceedings of Spatial Analysis and GEOmatics (SAGEO)* 2017 (Rouen, France, 2017), INSA de rouen, pp. 264–277. doi:10.5281/zenodo.1006677.

www.josis.org

- [28] PERALTA, V. Data Quality Evaluation in Data Integration Systems. phdthesis, Université de Versailles-Saint Quentin en Yvelines ; Université de la République d'Uruguay, 2006.
- [29] PEREIRA, R. H. M., SARAIVA, M., HERSZENHUT, D., BRAGA, C. K. V., AND CONWAY, M. W. r5r: Rapid realistic routing on multimodal transport networks with r5 in r. *Findings* (2021). doi:10.32866/001c.21262.
- [30] PORTA, S., CRUCITTI, P., AND LATORA, V. The Network Analysis of Urban Streets: A Primal Approach. *Environment and Planning B: Planning and Design 33*, 5 (2006), 705–725. doi:10.1068/b32045.
- [31] PUCHER, J., DILL, J., AND HANDY, S. Infrastructure, programs, and policies to increase bicycling: An international review. *Preventive Medicine* (2010).
- [32] RAIFER, M., TROILO, R., KOWATSCH, F., AUER, M., LOOS, L., MARX, S., PRZYBILL, K., FENDRICH, S., MOCNIK, F.-B., AND ZIPF, A. OSHDB: a framework for spatiotemporal analysis of OpenStreetMap history data. *Open Geospatial Data, Software and Standards* 4, 1 (2019), 3. doi:10.1186/s40965-019-0061-3.
- [33] SCHMIDL, M., NAVRATIL, G., AND GIANNOPOULOS, I. An approach to assess the effect of currentness of spatial data on routing quality. *AGILE: GIScience Series* 2 (2021), 1–12. doi:10.1234/56789.
- [34] SHAO, Z., ZHANG, Z., WEI, W., WANG, F., XU, Y., CAO, X., AND JENSEN, C. S. Decoupled Dynamic Spatial-Temporal Graph Neural Network for Traffic Forecasting. Tech. Rep. arXiv:2206.09112, arXiv, 2022. arXiv:2206.09112 [cs] type: article.
- [35] SVENSSON, A., JONSSON, L., AND HYDÉN, C. SEPARATION BETWEEN PEDESTRI-ANS AND BICYCLISTS. In TRB's 3rd Urban Street Symposium (2007).
- [36] VASSI, A., AND VLASTOS, T. A review and critical assessment of cycling infrastructures across Europe. *WIT Transactions on Ecology and The Environment* 191 (2014).
- [37] VYBORNOVA, A. Identifying and classifying gaps in the bicycle network of Copenhagen. Master's thesis, University of Copenhagen, 2021.
- [38] WANG, X., MA, Y., WANG, Y., JIN, W., WANG, X., TANG, J., JIA, C., AND YU, J. Traffic Flow Prediction via Spatial Temporal Graph Neural Network. In *Proceedings* of The Web Conference 2020 (New York, NY, USA, 2020), WWW '20, Association for Computing Machinery, pp. 1082–1092. doi:10.1145/3366423.3380186.
- [39] WU, H., CHENG, J., HUANG, S., KE, Y., LU, Y., AND XU, Y. Path problems in temporal graphs. *Proc. VLDB Endow.* 7, 9 (2014), 721–732. doi:10.14778/2732939.2732945.
- [40] WYSLING, L., AND PURVES, R. S. Where to improve cycling infrastructure? Assessing bicycle suitability and bikeability with open data in the city of Paris. *Transportation Research Interdisciplinary Perspectives* 15 (2022), 100648. doi:10.1016/j.trip.2022.100648.
- [41] YEBOAH, G., PORTO DE ALBUQUERQUE, J., TROILO, R., TREGONNING, G., PERERA, S., AHMED, S. A. K. S., AJISOLA, M., ALAM, O., AUJLA, N., AZAM, S. I., AZEEM, K., BAKIBINGA, P., CHEN, Y.-F., CHOUDHURY, N. N., DIGGLE, P. J., FAYEHUN, O., GILL, P., GRIFFITHS, F., HARRIS, B., IQBAL, R., KABARIA, C., ZIRABA, A. K., KHAN,

A. Z., KIBE, P., KISIA, L., KYOBUTUNGI, C., LILFORD, R. J., MADAN, J. J., MBAYA, N., MBERU, B., MOHAMED, S. F., MUIR, H., NAZISH, A., NJERI, A., ODUBANJO, O., OMIGBODUN, A., OSUH, M. E., OWOAJE, E., OYEBODE, O., PITIDIS, V., RAH-MAN, O., RIZVI, N., SARTORI, J., SMITH, S., TAIWO, O. J., ULBRICH, P., UTHMAN, O. A., WATSON, S. I., WILSON, R., AND YUSUF, R. Analysis of OpenStreetMap Data Quality at Different Stages of a Participatory Mapping Process: Evidence from Slums in Africa and Asia. *ISPRS International Journal of Geo-Information 10*, 4 (2021), 265. doi:10.3390/ijgi10040265.

- [42] YOON, J., BLUMER, A., AND LEE, K. An algorithm for modularity analysis of directed and weighted biological networks based on edge-betweenness centrality. *Bioinformatics* (2006). doi:10.1093/bioinformatics/btl533.
- [43] ZHAO, P., JIA, T., QIN, K., SHAN, J., AND JIAO, C. Statistical analysis on the evolution of OpenStreetMap road networks in Beijing. *Physica A: Statistical Mechanics and its Applications* 420 (2015), 59–72. doi:10.1016/j.physa.2014.10.076.